Reviewer Response

Reviewer #1:
Scientific Quality: Grade B (Very good)
Language Quality: Grade C (A great deal of language polishing)
Conclusion: Accept (General priority)
Specific Comments to Authors: The topic if the manuscript is highly interesting, i recommend acceptance of the manuscript after minor revision. Title of table 1 was repeated 2 times I recommend that authors need to rephrase manuscript.
Response: Thank the reviewers. We have checked and revised the full text according to the reviewer's comments.

Reviewer #2:
Scientific Quality: Grade B (Very good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Accept (General priority)
Specific Comments to Authors: it is a well written and documented case report
Response: Thank the reviewers. We have checked and revised the full text according to the reviewer's comments.